Author's response to reviews

Title: Improving performance in primary care through the meaningful use of electronic medical records: Test of an integrated structural model

Authors:

Louis Raymond (louis.raymond@uqtr.ca)
Guy Pare (guy.pare@hec.ca)
Ana Ortiz-de-Guinea (ana.ortiz-de-guinea@hec.ca)
Placide Poba-Nzaou (poba-nzaou.placide@uqam.ca)
Marie-Claude Trudel (marie-claude.trudel@hec.ca)
Josianne Marsan (josianne.marsan@sio.ulaval.ca)
Thomas Micheneau (thomas.micheneau@hec.ca)

Version: 2
Date: 3 November 2014

Author's response to reviews:

The title page has been modified so that the authors' names, affiliations, countries and email addresses now appear. Line numbers on each page have also been added as requested.